

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Harasztosi, Péter; Maurin, Laurent; Pál, Rozália; Revoltella, Debora; Van Der Wielen, Wouter

#### Working Paper Firm-level policy support during the crisis: So far, so good?

EIB Working Papers, No. 2022/01

**Provided in Cooperation with:** European Investment Bank (EIB), Luxembourg

*Suggested Citation:* Harasztosi, Péter; Maurin, Laurent; Pál, Rozália; Revoltella, Debora; Van Der Wielen, Wouter (2022) : Firm-level policy support during the crisis: So far, so good?, EIB Working Papers, No. 2022/01, ISBN 978-92-861-5210-8, European Investment Bank (EIB), Luxembourg, https://doi.org/10.2867/24308

This Version is available at: https://hdl.handle.net/10419/249917

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



### WWW.ECONSTOR.EU



# Firm-level policy support during the crisis:

So far, so good?



## Firm-level policy support during the crisis:

## So far, so good?

January 2022



Firm-level policy support during the crisis: So far, so good?

© European Investment Bank, 2022. EIB Working Paper 2022/01 January 2022

This is a publication of the EIB Economics Department. economics@eib.org www.eib.org/economics

#### About the Economics Department

The mission of the EIB Economics Department is to provide economic analyses and studies to support the Bank in its operations and in the definition of its positioning, strategy and policy. The department, a team of 40 economists, is headed by Director Debora Revoltella.

#### Authors

Péter Harasztosi, Laurent Maurin, Rozália Pál, Debora Revoltella and Wouter van der Wielen (EIB)

#### Disclaimer

The views expressed in this publication are those of the authors and do not necessarily reflect the position of the European Investment Bank.

EIB working papers are designed to facilitate the timely exchange of research findings. They are not subject to standard EIB copyediting or proofreading.

For further information on the EIB's activities, please consult our website, www.eib.org. You can also contact our InfoDesk, info@eib.org.

Published by the European Investment Bank. Printed on FSC® Paper.

## Firm-level policy support during the crisis: So far, so good?

#### Péter Harasztosi, Laurent Maurin, Rozália Pál, Debora Revoltella and Wouter van der Wielen\*

We use the 2021 vintage of the EIB Investment Survey (EIBIS) which contains a detailed set of questions regarding the nature of the policy support to firms during the COVID-19 crisis. Matched with hard data on the balance sheets and Profit and Loss (P&L) statements of corporations, the survey enables to disentangle the drivers of policy allotment and the impact of the policy support during the investment recovery.

First, we focus on the distribution of the policy support and show that it has been allotted mostly owing to the sales losses during the crisis, going to firms most affected during the crisis. We do not find evidence that the support was tilted towards firms already weak before the crisis.

Second, we show that the firms that have benefitted from the policy support tend to be more optimistic regarding their investment plans. The impact is especially pronounced for investment in digital technologies.

#### **Key Words**

EIBIS - COVID-19 Policy support in the EU - Economic rebound

#### J.E.L classification: E22, D22, H0

\* All European Investment Bank (EIB). p.harasztosi@eib.org, l.maurin@eib.org, r.pal@eib.org, d.revoltella@eib.org and w.vanderwielen@eib.org. The authors would like to thank Liza Archanskaia (EC), Sandra Philippen (ABN-AMRO), Christpoh Weiss (EIB) as well as participants to the EC Annual Research Conference (2021) for their comments and suggestions. The views expressed are those of the authors and do not necessarily reflect those of the European Investment Bank.

#### 1. Introduction

The COVID-19 crisis unfolded very differently from the two recent crises Europe endured, the Great Financial Crisis (GFC) and Sovereign Debt Crisis (SDC). While real GDP fell more than in the GFC, the economic effects of the pandemic unfolded differently. First, corporate profits did not decline that much (EIB, 2021). Second, bank lending spreads did not increase across countries (Altavilla et al, 2020), nor across borrower size (Andersson et al., 2021)<sup>1</sup>. Third, investment fell less compared to the investment in the GFC and the rise in unemployment was contained, thereby limiting the risks of scarring (Portès, 2020). Major disruptions in the European corporate ecosystem were avoided and corporate bankruptcy rates even declined during the crisis.

It is widely acknowledged that this positive outcome mostly resulted from the strength of the policy support deployed, by both monetary, financial and fiscal authorities (Djankov and Zhang, 2021). Monetary and financial supervisory policies have massively contributed to dampening the damaging impact of the crisis. The channels through which they have propagated are more conventional. They have been analysed at length in the literature even before the COVID-19 crisis, be it in the context of the GFC, SDC or the transition towards Basel III strengthened capital requirement.<sup>2</sup>

In this paper, we focus on another layer of policy support. We aim at analysing the nature and impact of the support deployed by the sovereigns and the EU on the corporate ecosystem at the firm level. This component has not benefited from so much attention, so far at least, mostly due to the lack of a homogenous dataset yet available, matching policy support and firm characteristics. In this paper, we use a very rich dataset to analyse the nature of this support in detail. Our empirical analysis is based on firm data drawn from the EIB Investment Survey (EIBIS). We focus on the policy support related questions in the 2021 vintage of EIBIS to illustrate the effectiveness of the support implemented in shielding corporates and fostering investment post-COVID-19.

Our analysis is related to the emerging debate on the possible side effects of the policy support. As the policy allotment was mostly unconditional, some concerns have been raised about the misallocation of public funds towards the maintenance of corporates that would otherwise have exited, even in the absence of crisis. If this is the case, misallocation of production factors would be more pronounced after the crisis and would weigh on the long-term prospects (Archarya et al., 2020).

We analyse the factors that can enable firms to recover stronger from the crisis, reaching higher productivity levels. We estimate linear and non-linear probability models to explain the

<sup>&</sup>lt;sup>1</sup> Andersson et al. (2021) show that adverse bank lending supply shocks such as those occurring during the GFC/SDC lower loan volumes and rise size spreads as the tightened loan supply has been more adverse for small corporations with limited market access.

<sup>&</sup>lt;sup>2</sup> See e.g., Kanngieser et al. (2019) on the macroeconomic impact of banks' capital adjustement, Ferrando et al. (2019) on the Outright Monetary Transactions (OMT) programme, Betz and De Santis (2021) on the Corporate Sector Purchase Programme (CSPP) and Andreeva and García-Posada (2020) on Targeted Longer-Term Refinancing Operations (TLTROS).

likelihood of getting policy support and its impacts on investment plans, controlling for the changes in demand at the sector level and the changes in the balance sheet structure. We show that the policy support has been allotted mostly owing to the sales losses during the crisis, going to the firms most affected. We do not find evidence that the support was tilted towards firms already weak before the crisis. We also find that the firms that benefitted from the policy support tend to be more optimistic regarding their investment plans. The impact is especially pronounced for investment in digital technologies. In some cases, it is strengthened by the recapitalisation that the policy support on the European economy precisely, we believe our results can answer part of the questions of this debate.

The rest of the paper is organised as follows. Section 2 reviews the main finding of the literature on policy support. Section 3 analyses the determinants of the allotment of the policy support. Section 4 analyses its impact on firms' investment plans. Section 5 provides concluding remarks.

#### 2. An overview of the impact of the policy support measures

To dampen the adverse economic impact of the COVID-19 pandemic, national governments implemented a broad set of measures to support the resilience of firms including public loans, state-backed loans guarantees, debt moratoria, job retention schemes (i.e. short-time work, furlough and wage subsidy schemes), tax and social security deferrals and relief, and (targeted) subsidies and grants. In Europe, the measures primarily aimed at supporting the resilience of firms, e.g. by assuring access to finance (via credit guarantees), temporarily reducing costs (via job retention schemes and tax deferrals) and direct support. Whereas the policy mix differed across countries and sectors, income and profit tax deferrals, loan guarantees and job retention schemes are among the most widely used measures. According to EIBIS 2021, the wide-ranging impact of the pandemic led to a broad coverage resulting in one in two businesses being subject to a measure. Joint use of multiple schemes was more limited, yet non-negligible, e.g. 30% of German firms made use of more than one support measure.<sup>3</sup>

While the support measures implemented in Europe are relatively similar, comparing them across countries is still complicated due to a variety of factors. First, the policy mix and parameters, related to the eligibility criteria, the support size or horizon, differ substantially. Second, the volumes of financial support initially set out in the programmes, the envelopes initially budgeted, differ from the amounts disbursed. Third, the COVID-19 policy support measures coincide with other accommodating policies affecting the corporate sector, e.g. the EU's Next Generation EU (NGEU), the ECB's pandemic emergency purchase programme (PEPP), and national support programmes for households as well as automatic stabilizers.

The analysis of the COVID-19 support measures can build on a vast pre-existing literature of microeconomic impact assessments. The impact of loan guarantee schemes at the firm level has been well documented for normal times. Loan guarantees offered by the US Small Business

<sup>&</sup>lt;sup>3</sup> Cross-country details are included in section 3.1.

Administration, for example, have been found to create 3 to 3.5 jobs per USD 1 million of loans (Brown and Earle, 2017). Similarly, EU evidence attributes loan guarantees with a significant positive impact on firms' size, revenues, employment, investment and survival (Asdrubali and Signore, 2015; Bertoni et al., 2018) and innovation (Bertoni et al., 2019; Brault and Signore, 2019). Martín-García and Morán Santor (2021) moreover find a significant impact of credit guarantees by the regional mutual guarantee institution on firm growth in the Spanish Region of Madrid. Country studies for Italy and the Czech Republic, on the other hand, do not find a significant impact (Dvouletý et al., 2019; D'Ignazio & Menon, 2020). Likewise, past subsidized loan programmes for SMEs have been found to have positive effects on job creation, investment and productivity in Bulgaria (Erhardt, 2017) and Hungary (Horvath & Lang, 2021, Endresz et al., 2015). Finally, firm-level evidence shows that job retention schemes prevent layoffs and safeguard firms' survival, see e.g. Hoffman and Schneck (2011), Cahuc et al. (2018), Lydon et al. (2019), Kopp and Siegenthaler (2019) and Guipponi and Landais (2020).

Although constrained by data availability limitations, early firm-level evidence is positive on the impact of national COVID-19 policy support measures. A selection of model-based simulation exercises have helped to inform policymakers from early on in the pandemic, such as Gourinchas et al. (2020, 2021), Lopez-Garcia (2020), Barnes et al. (2021), Blanco et al. (2021), Demmou et al. (2021a,b), Díez et al. (2021), and Ebeke et al. (2021), Maurin and Pal (2020). These studies have highlighted the potential of support measures to reduce liquidity shortfalls, bankruptcies, as well as output and employment losses relative to a no-policy scenario. Nevertheless, the true realized impact can only be gauged as detailed firm records become available.

More than one and a half years into the pandemic ex-post firm-level evidence is emerging.<sup>4</sup> Hadjibeyli et al. (2021), for example, perform a microsimulation exercise using French firmlevel data up to December 2020. The simulations show a 12 percentage points (pp.). lower increase in illiquidity and a 5.3 pp. lower increase in insolvencies when accounting for short-time work, direct subsidies and tax reliefs relative to a scenario without such policies.<sup>5</sup> Building on a similar, yet smaller database for 2020, Bureau et al. (2021) simulate a reduction from 60% to 47% in the share of firms facing a negative cash flow shock as a result of the French support measures (not including loan guarantees). France Stratégie (2021) extend the simulations to data up to March 2021, showing that the support measures (including loan guarantees) reduced the simulated share of firms facing a drop of the gross operating surplus of more than 25 value-added points by 13 pp. Alternatively, Drabancz et al. (2021) employ firm records up to December 2020 to provide causal evidence for Hungary's subsidized loan programmes, showing a 4% higher headcount in firms with five or more employees that received support. Lalinsky and Pal (2021) use firm-level data from Slovakia for March-June 2020 to investigate

<sup>&</sup>lt;sup>4</sup> In parallel to the evidence on European support measures, a vast literature assessing the US' loan guarantees under the Paycheck Protection Program (PPP) has emerged. Autor et al. (2020), Hubbard and Strain (2020), Granja et al. (2020), Doniger and Kay (2021) and Faulkender et al. (2021) all finding mild to strong evidence for the PPP boosting employment during the pandemic. Lopez and Speigel (2021) moreover find a marked positive effect of the program on small businesses' growth. Kapinos (2021a,b), nonetheless, draws a more heterogeneous picture of the program's impact.

<sup>&</sup>lt;sup>5</sup> The strong findings for liquidity are in line with corporate survey results showing that government fiscal support measures were effective in easing the liquidity needs of firms (De Santis et al., 2021).

government wage subsidies. They find significant drops in firms' probability of illiquidity (3.5%) and insolvency (3.5%) when granted support. Both studies moreover find stronger effects for smaller firms. In turn, this paper uses firm-level data for the EU, to gauge the impact of policy support on corporates' investment outlook.

In addition to the firm-level impact, public guarantee schemes are likely to have spurred bank lending. Casanova et al. (2021) present suggestive evidence that the aggregate bank lending to the private non-financial sector grew faster in countries with larger public guarantee schemes. Bank lending moreover was dependent on bank size and the condition of banks' IT systems (Core and De Marco, 2021), with larger and/or better IT equipped banks disbursing guaranteed loans faster. The importance of pre-existing lending relationships, as established for pre-crisis lending (Degryse and Ongena, 2005), was also confirmed for the lending operations during the COVID-19 pandemic.

The positive short-run impact of the policy support measures does not preclude medium-term risks for the sovereigns. While the pandemic-related loan guarantees have the benefit of spreading the exposure, the guarantees issued in response to COVID-19 tend to be concentrated among the most vulnerable firms and the hardest-hit sectors. For example, recent firm-level evidence from Italy finds that financially fragile firms – in particular smaller, less liquid, more leveraged firms and/or firms are more likely to have received public guaranteed loans during COVID-19 (Core and De Marco, 2021). Interestingly, firm-level evidence for four other EU countries (Croatia, Finland, Slovakia and Slovenia) suggest that this does not hold for employment subsidies and direct subsidies and supports have been distributed towards medium productive firms, and only marginally towards 'zombies'. (Bighelli et al., 2021). Moreover, an initial assessment of bank lending in Austria suggests that increased lending has contributed to the stabilization by providing liquidity for the economy at no additional risks (Kaniovski et al., 2021). Finally, the results of our EU-wide analysis confirm that the lion's share of firms benefitting from support were in good pre-COVID financial health.

In sum, national COVID-19 policy support measures have played an important role in stabilizing the economy and bank lending during the COVID-19 pandemic. Preliminary firm-level evidence show a decisive role for the support measures in limiting insolvencies and safeguarding employment. The public support schemes that were introduced in response to the COVID-19 pandemic – in particular state-backed loan guarantees –, however, also constitute sizeable contingent liabilities for governments, therefore raising concerns about the potential medium-term risks for sovereigns. Already, the decline in output combined with increased expenditures caused a strong increase in government debt. The COVID-19 crisis erased fiscal consolidation efforts of previous years and in 2021, debt levels in the EU were above those in 2014. Moreover, GDP declined more in Member States with high debt levels, driven mainly by higher exposure to more affected sectors, like the hotel and restaurant services and transportation sectors.

In the two next sections, we rely on the 6<sup>th</sup> vintage of the European Investment Bank Investment Survey, the EIBIS2021. The annual survey contains information on more than 12000 non-

financial firms in the EU.<sup>6</sup> The sampling is representative at the country and sector level for the four sectors considered, manufacturing, services, construction and infrastructure (see Brutscher et al. 2020 for details). The 6<sup>th</sup> vintage contains several questions related to policy support. We match the answer of the firms with various pre-COVID-19 balance sheets and P&L characteristics to analyse the allotment of the policy support in Section 3. The survey also contains information about investment plans, their volume as well as their nature, therefore, enabling the analysis of the policy support in Section 4.

#### 3. Allotment of the policy support

In this section, we focus on the nature of the policy support deployed across European economies and investigate its targeted nature using the information collected during the  $6^{th}$  vintage of the EIBIS.

It is important to note upfront that EIBIS2021 survey information does not enable to distinguish whether the absence of policy support reflects a voluntary choice or non–eligibility.<sup>7</sup> Indeed, the recourse to policy support needs both and it is likely that the choice not to apply is indicative of certain firm characteristics or strategies. For example, France Stratégie (2021b) shows that the intensity of recourse decreases with business size, that the recourse to several measures is not systematic, and the non-use appears to be largely voluntary. Besides the eligibility to participate in the programmes, technical factors also matter for their deployment. The operational design of the programmes, the setup of the administrative processes have been used to explain the low take-up rate at the beginning of the crisis. Two key bank characteristics facilitated loan disbursement: size and information technology. Core and de Marco (2021) show that these factors were important because of the high volume of online applications and low-interest margins on guaranteed lending. Pre-existing relationships mattered for the allocation of guaranteed credit, as banks lent more in their core markets and where they have a larger local market share.

In the following, we focus on the effective policy support.

#### 3.1. Nature and intensity of the policy support deployed in EU economies

In the EIBIS, four types of support are considered: (1) New subsidized or guaranteed credits (e.g. loan, overdraft or credit card from a bank or other finance provider) that will need to be paid back in the future but may have preferential or reduced interest rates and/or an extended repayment plan (2) Deferral of payments which still leave a liability to be paid by the company in the future (e.g. deferral of tax payments, deferral of rent or mortgage for commercial property, suspension of interest payments), (3) subsidies or any other type of financial support

<sup>&</sup>lt;sup>6</sup> EIBIS also surveys close to 1000 firms in the US and UK. Those are not used in this analysis.

<sup>&</sup>lt;sup>7</sup> Such information is available in the 2021 EIBIS "Add-on module", albeit for a more limited number of companies, around 1000. The Add-On Module targeted non-financial SMEs operating in manufacturing and services to gain additional insights on the impact of Covid-19, climate change and digitalisation on companies investment needs across the EU.

that the company will not have to pay back in the future, a type of support that comprises job retention policies and (4) any other type of financial support.

**Figure 1** reports the share of companies that have benefitted from policy support across the EU. 44% of EU corporates did not get support, a ratio varying from 22% in Luxembourg to 61% in Estonia. This ratio is somewhat lower in Southern economies.



Figure 1 Occurrence of the policy support across European economies (% of recipient firm)

Overall, in Europe, 56% of firms received support via at least one specific policy. The majority of the firms received only one type of support. Around a third of companies that received support benefitted from two types of policies. This represents 12% of firms at the EU level. 4% of firms received all three types of support. Among types of policy support, subsidized financial support (type 3) is the most common, used by 36% of the firms. The ratio was even higher in the Northern and Western economies and in the Central and Eastern economies. A similar share of firms, 16-17%, benefitted from the deferral of payments or credit support to be paid back.<sup>8</sup>

Turning to the different layers of policy support, **Figure 2** plots the share of corporations benefitting from two types of support by country. In the left panel, the intensity of new subsidized or guaranteed credits (type 1) is correlated to that of subsidies or any other type of financial support (type 3). The negative relationship suggests that these two types of policies have been substitutes to one another: countries where firms benefit more from subsidies or any other type of financial support are countries where firms benefit less from new subsidized or guaranteed credits. The negative correlation amounts to -31%. However, it is important to notice that the negative correlation is strongly driven by a set of countries, such as Spain, Greece, Portugal, Italy and France, where the subsidised and guarantied loans were more

**Source**: Calculations based on the EIBIS 2021. **Note**: % of firm having benefitted from at least one type of support. The colour reflects the region in which the economy is located: Red indicates Central and Eastern economies, Green indicates Southern economies and Orange indicates Northern and Western economies.

<sup>&</sup>lt;sup>8</sup> Type 4, "any other type of support", is reported by a small share of companies and is very heterogeneous across countries. Therefore, it is not considered in isolation in this paper.

intensively used. These are also those EU countries that were particularly hit by the crisis, mainly due to higher exposure to the tourism and travel sectors. When eliminating from the series these particular countries, the correlation between type 1 and type 2 policy support turns positive 14%.

In the figure on the right, deferral of payments (type 2) is correlated with subsidies or any other type of financial support not to be paid back (type 3). With a correlation of 30%, the allotment of both policies is positively linked; countries where firms benefit more from one policy also tend to benefit more from the other. This suggests that most of the measures that fall under the category subsidies or any other type of financial support (type 3) are likely to be labour support policies. The allotment of this support has been associated with the deferral of social contributions or tax payments.



*Source*: Calculations based on the EIBIS 2021. *Note*: See note Figure 1 for an explanation of the colours.

#### 3.2. The support went to the firms most affected by the crisis

**Figure 3** considers sector asymmetries using two different breakdowns. First, in the left panel, we look at the share of firms with policy support against the share of firms with severe sales loss, more than 25% for 12 broad sectors.<sup>9</sup> The panel confirms that services comprise some of the sectors most hit, such as hotel and restaurant, and some not or positively affected, such as IT and telecommunications. The positive correlation with the prevalence of policy support suggests that, across sectors, the support was much linked to the change in activity: the stronger the decline in turnover in the sector, the higher the intensity of the policy support. With an R-square of 76%, there is a strong relationship. When types of policy supports are investigated separately, financial subsidies (type 3) shows the strongest relationship with losses. This again relates to the composition of this policy component temporary unemployment directly linked to activity.

<sup>&</sup>lt;sup>9</sup> While the EIBIS sampling is not designed to be representative of this twelve sectors, each sector is populated by 350 firms at least. In fact, the four sectors decomposition featured in the EIBIS provides only an aggregated view on the impact of the crisis.



#### Figure 3 Determinant of the allotment of policy support (% of firms)

**Source:** Calculations based on EIBIS 2021. Note: Any type of policy support is considered simultaneously. Firm sizes: (S)-micro and small firms (with between 5 and 50 employees), (M) – medim sized firms (with between 51 and 250 employees), (L) – large firms (with more than 250 employees).

Looking at the right-hand panel of **Figure 3**, we use the breakdown of four sectors together with a breakdown by firm size categories. Firms in the manufacturing and services sector are positioned in the right, implying that they have been more affected. In the same sector, smaller firms were more affected and therefore more likely to get supported. For each of the four sectors considered separately, the proportion of allotment to smaller companies is higher than for larger companies. This seems to reflect the fact, as discussed above, that the smaller firms have reported higher sales losses (Gourinchas et al., 2021). Smaller firms are positioned to the right of their larger counterparts. During the COVID-19 crisis, small firms were more likely to have suffered large sales losses (above 25%) compared to large firms, respectively 29.1% vs 9.1% in the manufacturing sector, 34.5% vs 28.9% in the services sector, 17.6% vs 1.2% in the construction sector and 25.9% vs 16.1% in the infrastructure sector. The difference across sizes is larger in the construction and manufacturing sectors.

Overall, the support went mainly to firms reporting larger sales drops. **Figure 4** reports the distribution of the change in sales for firms that received the support and those that did not. The two distributions are different. Firms that received support have a distribution tilted to the left. For the support, the mode of the distribution corresponds to a decline between 0 and 25% while the unsupported have a mode around 0. Hence, supported companies have reported higher declines in sales.

In **Figure 5**, we report the share of companies that received each type of support, separately for the three policies. Again, the larger the sales loss, the higher the intensity of the support has been. The difference is especially strong for the allotment of subsidies and financial support as in this case, the difference between zero and major sales loss reaches 30%. This is consistent with the nature of this policy support, mostly comprising labour support policies, and therefore even more tied to demand. Conversely, subsidized or guaranteed credits tend to be less targeted.

This could suggest that corporates took the opportunity of cheap credit availability to insure themselves against possible liquidity shortfalls.



Figure 5 Policy allotment conditioned on sales losses



**Source:** Calculations based on EIBIS 2021. **Note**: The y-axis indicate the share of firms reporting a change in sales indicated on the x-axis.

*Source:* Calculations based on the EIBIS 2021. *Note:* The y-axis indicate the proportion of firms having benefitted from the support. Minor (Major) change corresponds to less (more) than 25%.

To better estimate the dependence of the policy allotment on the change in sales during the COVID-19 crisis, we propose the following linear probability model:

$$q_{i,c,s}^{k} = \alpha Sales_{i} + \Gamma \boldsymbol{\theta}_{i} + \varepsilon_{i}$$
 EQ. 1

where q is the answer to the question on the policy support in the EIBIS. It is a dummy variable indicating whether the firm received the policy support. Since several types of policy support are considered, the dummy is indexed by k, the type of support that can be of three types. The variable *Sales* is categorical variable indicating the change in sales as reported in the survey. We use this variable either as an indicator for negative change or break down the sales loss indicator into brackets.  $\theta_i$  denote a vector of dummy variables controlling for country, sector size and firm age. Consistently with the right-hand panel of Figure 3, the four EIBIS sectors are considered in the regression, since those are by construction covered representatively. Again, consistently with Figure 3, the three size groups, micro and small, medium and large are considered. The results are presented in Table 1 and Figure 6.

	 				<u> </u>	v	
	(1)		(2)		(3)	(4)	(5)
	Any p	olic	y support	ş	Subsidized or guaranteed credit	Payment deferral	Subsidies or any other type of financial support
Sales loss dummy	0.216***						
	[0.009]						
Sales loss categories							
less than 25%			0.165***		0.018**	0.068***	0.152***
			[0.011]		[0.007]	[0.009]	[0.011]
between 25 to 50%			0.272***		0.055***	0.095***	0.255***
			[0.013]		[0.010]	[0.011]	[0.013]
more than 50%			0.302***		0.073***	0.183***	0.280***
			[0.018]		[0.015]	[0.018]	[0.019]
Observations	10,929	•	10,929	•	10,929	10,929	10,929
R-squared	0.136		0.142		0.150	0.085	0.173
Firm controls	yes		yes		yes	yes	yes
Country FE	yes		yes		yes	yes	yes
Sector FE	yes		yes		yes	yes	yes

#### Table 1 Factors explaining the likelihood of getting policy support

**Source:** Authors' estimations based on EIBIS21. **Note:** Linear Probability Model estimated with firm size dummies (small, medium and large), sectors (as in Figure 3) and firm age dummies: less than 2 years; 2 years to less than 5 years; 5 years to less than 10 years; 10 years to less than 20 years; 20 years or more. Constant not reported. Robust standard errors in brackets, \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1.

In **Table 1**, the first two columns relate to any type of policy support while columns 3 to 5 analyse separately each policy layer. In the first estimation, the likelihood of getting policy support of any type is explained by the dummy reporting of whether firms have recorded sales losses or not. The coefficient is positive and highly significant. It shows that recording a decline in sales increases the probability to be supported by 21 pp. The equations reported in columns 2 to 5 consider a finer breakdown of sales losses, by brackets of 25%. The conclusions drawn from **Figure 4** and **Figure 5** are confirmed. The impact of sales loss is always significant, for any type of support, considered together or separately. Furthermore, it increases in intensity with the magnitude of the decline. The higher the loss, the more likely it is that the firm will get support. This is true for the policies in general. However, it is more pronounced for subsidies and other policy support, a component that includes labour support more linked to sales drops. Conversely, this is less conducive for subsidized and guaranteed loans.

**Figure 6** reports the coefficients attached to each decline. Focusing on the comparison of policy support measures, the impact of sales losses on the probability of getting the support is more linked to the magnitude of the loss for subsidies and other types of support. Clearly, the higher the losses the stronger the likelihood of getting the support. It should be stressed that this result is obtained after controlling for sector and country fixed effects. This policy appears to be linked to the actual change in the activity mostly. Conversely, the intensity of the change in sales is less determinant for subsidized or guaranteed loans.

**Figure 6 Impact of sales loss on the likelihood of getting the policy support** (*Elasticity, change in the probability, pp.*)



**Source:** Authors' estimations based on EIBIS 2021 and ORBIS. **Note:** the plain bar indicates the coefficient estimated in **Table 1** (columns 2-5). The vertical line indicates its 95% confidence interval. The x-axis indicates the change in sales reported by the company.

#### 3.3. The allotment is mostly unrelated to pre-crisis weakness

We now turn our focus to the discriminatory power of firm-specific characteristics in getting policy support. In particular, we estimate the probability of companies getting the support depending on their status before the crisis. After taking into consideration the impact of sales loss, we investigate the other factors likely to influence policy allotment. We consider factors related to the firm characteristics, its activity, its balance sheet structure or profit and loss accounts. For this, we estimate a linear probability regression, EQ2. Separate models are estimated for each pre-crisis firm characteristic, each controlling for the country, sector, size, age and sales decline.

$$q_{i,c,s}^{k} = \alpha Sales_{i} + \beta X_{i} + \Gamma \theta_{i} + \varepsilon_{i}$$
 EQ2

q is the answer to the question on the policy support in the survey. EQ2 differs from EQ1 as, in addition to the change in sales, it also investigates the role of further important firm characteristics ( $X_i$ ). The set of control variables are the same as in EQ1.

We first estimate the impact of the following firm characteristics: productivity (measured as value-added per employee), investment ratio (measured as the annual growth rate in fixed assets), being digital (implementing at least one type of digital technology), and being an exporter. We then analyse the impact of the following financial indicators: liquidity ratio (cash and equivalent over total assets), return on assets, distressed status (when the interest coverage ratio was below 1 pre-COVID-19), indebtedness (first and last decile of the financial leverage), registering losses in the year before the crisis, and finance constrained (rejected or discouraged for external financing).

To report the results of EQ2, **Figure 7** plots the change in the predicted probability of getting the policy support (of any type), while policy specific results are displayed in the Appendix. When the characteristic is binary, the presence or absence is reported as yes or no. When it is

continuous, high relates to being in the last decile and low in the first decile, while we also plot median for productivity.

Beyond the control variables, few factors appear to matter. Above all, the firm's productivity level appears as the main discriminant factor. Firms with low pre-COVID-19 productivity are significantly more likely to get policy support than firms with high-level productivity. Being an exporter also significantly matters, albeit to a lesser extent. Digital firms are more likely to get the support, although the difference to non-digital firms is not significant at the 95% confidence level. The investment ratio before the crisis does not seem to have a predictive impact on obtaining the support.



**Figure 7** Predicted probability of getting the policy support across firm characteristics **pre-COVID** (predicted probability controlled for sales loss, %)

Source: Authors' estimations based on the EIBIS2021 matched with the ORBIS database. Note: The vertical line reports the 95% interval confidence of the conditional probability of getting the support (see EQ2). Two overlapping lines indicate that the factor does not alter significantly the probability. Red bars indicate when the characteristic is statistically discriminant at the 95% confidence level.

Turning to the financial indicators, only firms with low liquidity, those with smaller cash buffers, are more likely to get policy support.<sup>10</sup> While firms with low return on assets and finance constrained firms are more likely to get support, the difference is not significant. This result suggests that the primary goal of the policy support, avoiding a liquidity dry-out and freezing the corporate ecosystem, was reached. Overall, we do not find evidence for Europe

<sup>&</sup>lt;sup>10</sup> As an alternative exercise, we checked also the correlation of sales loss during th pandemic with pre-covid firm caracteristics and financial indicators. Results show that the pre-crisis financial performance is uncorrelated with the sales loss, while some firm characteristics, such as high productivity and being digital has lower probability of losses.

that the support was tilted towards firms already weak before the crisis, such as financially distressed, firms registering losses, or highly indebted.<sup>11</sup>

#### 4. The policies' effectiveness

In this section, we analyse the role played by policies in supporting the investment recovery. Figure 8 plots the percentage of firms planning to invest more in the current financial year depending on the sale loss they have recorded over the COVID-19 period, distinguishing firms that have benefitted from the support and those that have not. The share of firms planning to invest more tend to increase when sales losses become lower. Moreover, comparing across groups of firms, for the same level of losses, those that have been supported plan to raise investment by more. The difference is especially pronounced for large sales losses. Figure 9 plots the share of firms that have increased their recourse to external finance, debt or equity, depending on whether they have benefitted from policy support. The balance sheet expansion is stronger for those that have been supported. In case of debt, part of the difference reflects the recourse to subsidized loans or guarantees and is therefore accounting the support. This is not the case for equity.

Figure 8 Investment plans conditional on Figure 9 Policy support and balance COVID-19 sales losses and policy support (% respondent)



Source: computations based on the EIBIS 2021. Note: the xaxis reflects the sale losses reported by the company. The yaxis reports the percentage of firms surveyed that plan to raise investment in the current financial year.





Source: computations based on the EIBIS 2021.

#### 4.1. Some evidence that the policy support contributes to the investment rebound

In the EIBIS 2021, several questions are related to investment, either planned or achieved. The question we use in this section relates to the change in financial plans, whether firms revised

<sup>&</sup>lt;sup>11</sup> Bighelli et al. (2021) reach similar conclusions. The authors show that Covid-19 support was distributed rather efficiently as government subsidies were distributed towards medium productive firms, and only marginally towards the undeserving 'zombies'.

their investment plans due to the COVID-19 pandemic, and if so, in which direction, upwards or downwards. We consider the firms that plan to invest more in the current financial year, i.e. those reported in Figure 8.

We aim isolating the channel through which the policy impacts investment while controlling for the characteristics that influence the decision to invest. For this purpose, we estimate the following equation:

$$q_{i,c,s} = \alpha. Sales_i + \beta. Pol_i^k + \gamma. Sales_i \times Pol_i^k + Z_i + \Gamma \boldsymbol{\theta}_i + \varepsilon_i$$
 EQ. 3

As before, q is the dummy variable reflecting the firm's answer to the survey regarding investments. Sales is set of dummy variables indicating the decline in sales the firm reported by 25% gaps. Pol indicates that the firm has benefitted from at least one policy support measure. Each dummy takes the value one when the answer is positive and zero otherwise. EQ 3 also includes an interaction term between policy access and the sales decline indicators. This allows to assess whether firms that suffered larger losses (any were more likely to access the support) reacted to the policy support relatively less or more. Z is a set of real and financial characteristic. Labour productivity is always incorporated in the equations, as a standard determinant of investment.<sup>12</sup> It is considered before the COVID-19 crisis to minimise the risk of being impacted by the decline in sales during the crisis. Besides, we alternatively consider the financial leverage, firms in financial distress, and the capital ratio. Financial distress is the dummy variable considered as before while the two other characteristics are taken as continuous variables. Financial leverage is defined as the sum of loans and long-term debt divided by fixed assets. The capital ratio is defined as shareholder's funds over total assets.<sup>13</sup> We expect a positive coefficient, considering that those with a financial leverage has access to external financing in forms of loans are able to invest more than those relying just on their internal sources (no financial leverage). Nevertheless, above a treshold, financial leverage might indicate debt overhang and in therefore have a negative impact on investment plans. As creditor risk increases with financial distress, this is also expected to have a negative coefficient. Conversely, for the same reasons, the capital ratio is expected to have a positive impact. As before, the regressions include dummies to account for the sector, size, age and the country effects. Standard errors are clustered at the firm level.

**Table 2** highlights the result that firms which benefitted from policy support are more likely to increase investment in 2021. First, in all the estimations, the policy allotment has a positive impact. Conversely, recording large sales losses has a negative and always significant impact. The positive coefficient on the interaction effect Sales  $\times$  Policy indicates that for the same decline in losses, investment prospects are more positive for firms that have benefitted from policy support than for those that have not. This result may suggest that the support has cushioned firms balance sheets and enabled them to rebound stronger. While positive across estimations, the effect is mostly not significant, however. Finally, the firm's characteristics and financial indicators not matter significantly. The coefficient on sales remains relatively

<sup>&</sup>lt;sup>12</sup> Estimations results are not sensitive to the inclusion of productivity in the set of regressors.

<sup>&</sup>lt;sup>13</sup> All these variables are winsorized to 3% and 97%.

constant across estimations, which suggest that the sale shock is relatively orthogonal to firm characteristics, consistently with the findings of Section 3. Firms reporting a sales loss of more than 25% are 6 to 9 pp less likely to increase investment.

imanciai ycai									
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	_		_	_		_	_	_	_
Policy support	0.011	0.024**	0.012	0.006	0.023**	0.022	0.010	-0.005	0.020
	[0.011]	[0.011]	[0.012]	[0.013]	[0.011]	[0.015]	[0.016]	[0.018]	[0.015]
Sales loss									
less than 25%		-0.027**	-0.022	-0.023	-0.025*	-0.024	-0.019	-0.034	-0.023
		[0.013]	[0.013]	[0.015]	[0.013]	[0.019]	[0.020]	[0.023]	[0.019]
between 25 to 50%		-0.065***	-0.056***	-0.040**	-0.063***	-0.088***	-0.085***	-0.084**	-0.087***
		[0.015]	[0.017]	[0.019]	[0.016]	[0.026]	[0.027]	[0.034]	[0.026]
more than 50%		-0.112***	-0.101***	-0.108***	-0.111***	-0.092**	-0.065	-0.069	-0.087**
		[0.022]	[0.024]	[0.029]	[0.023]	[0.040]	[0.045]	[0.055]	[0.041]
Policy support X						_	_	_	_
sales loss less than 25%						-0.006	-0.005	0.019	-0.002
						[0.025]	[0.027]	[0.030]	[0.026]
sales loss bw. 25 to 50%						0.032	0.041	0.063	0.035
						[0.032]	[0.034]	[0.041]	[0.032]
sales loss more than 50%						-0.027	-0.048	-0.046	-0.030
						[0.046]	[0.051]	[0.063]	[0.047]
Pre covid productivity	<b>•</b> 0.000	0.001	0.004	0.001	0.001	0.001	0.004	0.001	0.001
ric-covia productivity	0.000 [0.007]	-0.001	-0.004	100.001	-0.001	-0.001	-0.004 [0.008]	-0.001	-0.001 [0.007]
Financial Leverage	[0.007]	[0.007]		[0.009]	[0.007]	[0.007]		[0.009]	[0.007]
Tillahelai Leverage			0.008 [0.005]				0.008 [0.005]		
Firm in distress			[0.005]	0.006			[0.005]	0.006	
1 mm m distress				[0.017]				[0.017]	
Capital ratio				[0.017]	-0.010			[0.017]	-0.009
Cuphur futio					[0.020]				[0.020]
Observations	8.819	8.819	7,792	6.089	8.541	8.819	7,792	6.089	8.541
R-squared	0.025	0.028	0.028	0.024	0.027	0.028	0.029	0.024	0.027
Firm controls	yes	yes	yes	yes	yes	yes	yes	yes	yes
Country FE	yes	yes	yes	yes	yes	yes	yes	yes	yes
Sector FE	yes	yes	yes	yes	yes	yes	yes	yes	yes

 Table 2 Factors explaining the likelihood of increasing investment in the current financial year

**Source:** Authors' estimations based on EIBIS21. Note: Linear Probability Model estimated with firm size dummies (small, medium and large), sectors (as in Figure 3) and firm age dummies: Less than 2 years; 2 years to less than 5 years; 5 years to less than 10 years; 10 years to less than 20 years; 20 years or more. Constant not reported. Robust standard errors in brackets, \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1.

#### 4.2. Some evidence that the policy support has enabled financial expansion

We turn to the analysis of the policy support on the liability side of corporate balance sheets. We focus on whether the company has raised new equity from current stakeholders or not. We estimate EQ3 for the answers to this question.<sup>14</sup>

<sup>&</sup>lt;sup>14</sup> On equity, we consolidate two answers from Q68. Raising new equity (from the market, from external private investors) and raising new equity from current owners. We consider these two possibilities jointly.

The results are reported in **Table 3**. Across all the estimations, the policy support raises the likelihood of increasing the equity base, an effect always significant at a 1% confidence level. Somewhat surprisingly, sales losses also raise the probability of increasing the equity base. The conjunction of these two effects may suggest that recapitalisation needs resulting from large losses become more likely with the policy allotment. Getting it would increase the probability of crowd-in equity investors. Such interpretation is somewhat supported by the estimated impact of firm characteristics. The higher the financial leverage and the lower the capital ratio pre-COVID19, the more likely the increase in the equity base. Hence, the change in the financial structure possibly corrects balance sheet weakness.<sup>15</sup>

Policy support       0.024***       0.016***       0.014**       0.017***       0.013**       0.022***       0.019***       0.022***       0.019***       0.022***       0.019***       0.022***       0.019***       0.022***       0.019***       0.022***       0.019***       0.022***       0.019***       0.022***       0.019***       0.022***       0.019***       0.022***       0.019***       0.022***       0.019***       0.022***       0.019***       0.022***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.019***       0.012       0.016*       0.012       0.016*       0.012       0.016*       0.019***       0.012       0.016***       0.019***       0.012       0.016***       0.019***       0.011       0.009       0.0010
Policy support       0.024***       0.016***       0.014**       0.017***       0.013**       0.022***       0.019***       0.019***         [0.005]       [0.006]       [0.006]       [0.006]       [0.006]       [0.007]       [0.007]       [0.008]       [0.007]         Sales loss       less than 25%       0.008       0.009       0.006       0.009       0.016*       0.015*       0.012       0.016*         between 25 to 50%       0.045***       0.043***       0.036***       0.046***       0.058***       0.060***       0.039**       0.061***         more than 50%       0.068***       0.065***       0.052***       0.063***       0.053**       0.051*       0.073       [0.07]
$ \begin{bmatrix} 0.005 \end{bmatrix} \begin{bmatrix} 0.006 \end{bmatrix} \begin{bmatrix} 0.006 \end{bmatrix} \begin{bmatrix} 0.006 \end{bmatrix} \begin{bmatrix} 0.006 \end{bmatrix} \begin{bmatrix} 0.007 \end{bmatrix} \begin{bmatrix} 0.007 \end{bmatrix} \begin{bmatrix} 0.007 \end{bmatrix} \begin{bmatrix} 0.008 \end{bmatrix} \begin{bmatrix} 0.007 \end{bmatrix} \\ \begin{bmatrix} 0.008 \end{bmatrix} \begin{bmatrix} 0.007 \end{bmatrix} \\ \begin{bmatrix} 0.008 \end{bmatrix} \begin{bmatrix} 0.009 \end{bmatrix} \\ \begin{bmatrix} 0.006 \end{bmatrix} \begin{bmatrix} 0.009 \end{bmatrix} \\ \begin{bmatrix} 0.006 \end{bmatrix} \\ \begin{bmatrix} 0.009 \end{bmatrix} \\ \begin{bmatrix} 0.009 \end{bmatrix} \\ \begin{bmatrix} 0.009 \end{bmatrix} \\ \begin{bmatrix} 0.006 \end{bmatrix} \\ \begin{bmatrix} 0.009 \end{bmatrix} \\ \begin{bmatrix} 0.016 \end{bmatrix} \\ \begin{bmatrix} 0.017 \end{bmatrix} \\ \begin{bmatrix} 0$
Sales loss         0.008         0.009         0.006         0.009         0.016*         0.015*         0.012         0.016*           less than 25%         [0.006]         [0.006]         [0.007]         [0.006]         [0.009]         [0.009]         [0.010]         [0.009]           between 25 to 50% <b>0.043*** 0.036*** 0.046*** 0.058*** 0.060*** 0.039** 0.061***</b> [0.009]         [0.010]         [0.011]         [0.010]         [0.016]         [0.017]         [0.019]         [0.017]           more than 50% <b>0.068*** 0.052*** 0.063*** 0.058** 0.051* 0.073* 0.051*</b>
less than 25%       0.008       0.009       0.006       0.009       0.016*       0.015*       0.012       0.016*         [0.006]       [0.006]       [0.007]       [0.006]       [0.009]       [0.009]       [0.010]       [0.009]         between 25 to 50%       0.045***       0.043***       0.036***       0.046***       0.058***       0.060***       0.039**       0.061***         more than 50%       0.068***       0.065***       0.052***       0.063***       0.053**       0.051*       0.073*       0.051*
[0.006]         [0.006]         [0.007]         [0.006]         [0.009]         [0.009]         [0.010]         [0.009]           between 25 to 50%         0.045***         0.036***         0.046***         0.058***         0.060***         0.039**         0.061***           more than 50%         0.066***         0.065***         0.063***         0.063***         0.058**         0.051*         0.073*         0.051*
between 25 to 50%         0.043***         0.036***         0.046***         0.058***         0.060***         0.039**         0.061***           more than 50%         0.068***         0.065***         0.063***         0.063***         0.058**         0.061**
[0.009]         [0.010]         [0.011]         [0.016]         [0.017]         [0.019]         [0.017]           more than 50%         0.068***         0.065***         0.063***         0.063***         0.058**         0.051*         0.073*         0.051*
more than 50% 0.068*** 0.065*** 0.052*** 0.063*** 0.058** 0.051* 0.073* 0.051*
[0.016] $[0.016]$ $[0.019]$ $[0.016]$ $[0.028]$ $[0.029]$ $[0.038]$ $[0.028]$
Policy support X
sales loss less than 25% -0.014 -0.010 -0.010 -0.013
[0.012] [0.012] [0.014] [0.012]
sales loss bw. 25 to 50% -0.025 -0.025 -0.023
[0.020] [0.021] [0.023] [0.020]
sales loss more than 50% 0.010 0.016 -0.029 0.013
[0.032] [0.033] [0.042] [0.032]
Pre-covid productivity -0.015*** -0.014*** -0.015*** -0.012** -0.012*** -0.014*** -0.015*** -0.012*** -0.012***
[0.004] $[0.004]$ $[0.005]$ $[0.004]$ $[0.004]$ $[0.004]$ $[0.004]$ $[0.004]$ $[0.005]$ $[0.004]$
Financial leverage 0.004* 0.005*
[0.003] [0.003]
Firm in distress 0.051*** 0.051
[0.010] [0.010]
Capital ratio -0.050*** -0.050***
[0.011] [0.011]
Observations         8,819         7,792         6,089         8,541         8,819         7,792         6,089         8,541
R-squared 0.036 0.042 0.035 0.053 0.044 0.042 0.035 0.053 0.045
Firm controls yes yes yes yes yes yes yes yes
Country FE yes yes yes yes yes yes yes
Sector FE yes yes yes yes yes yes yes yes

**Table 3 Factors explaining the recapitalisation** (*firms that have raised new equity*)

**Source:** Authors' estimations based on EIBIS21. **Note:** Linear Probability Model estimated with firm size dummies (small, medium and large), sectors (as in Figure 3) and firm age dummies: Less than 2 years; 2 years to less than 5 years; 5 years to less than 10 years; 10 years to less than 20 years; 20 years or more. Constant not reported. Robust standard errors in brackets, \*\*\* p < 0.01, \*\*p < 0.05, \*p < 0.1.

<sup>&</sup>lt;sup>15</sup> See Maurin and Pal (2020) or Carletti et al. (2020) for the need to increase the capital base of corporates after the sharp fall in profits during the Covid-19 crisis.

#### 4.3. Some evidence that the policy support fastens firms digitalisation

In this section we focus on the question that relates to the investment response taken as a response to the COVID-19 pandemic, whether firms have taken any actions or made investments to (1) develop new products, services or processes, (2) transform their supply chain (3) become more digital. While we do not find conclusive and significant impact of the policy support in terms of developing new products, services or processes and transforming their supply chain, there is much clear reaction of firms in terms of digital transformation. We focus on firms that have taken action to become more digital while estimations of the two other type of investments are presented in appendix. The crisis has strengthened the case for European firms to digitalise, deeper and faster. Therefore as a last question, we consider the role of the policy support and the strengthening of the equity base in the digital transition. We estimate the following equation with or without the indication on equity injection:

$$q_{i,c,s} = \alpha. Sales_i + \beta. Pol_i^k + \gamma. Sales_i \times Pol_i^k + \phi. Equity_i + Z_i + \Gamma \theta_i + \varepsilon_i \qquad \text{EQ. 4}$$

In EQ3, Equity relates to the question on financial expansion, whether firms have raised equity, other explanatory variables are defined as in previous equations.

**Table 4** shows the results explaining the likelihood to increase digitalisation. Interestingly, in all the estimations, sales losses hurt digitalisation, reducing the likelihood to digitalise more by 5 to 9 pp. However, the effect is compensated by the policy allotment. On the one hand, firms that received policy support are unconditionally more likely to digitalise more, by 5 to 6 pp. On the other hand, firms that received the policy support while suffering large sale losses, are more likely to digitalise than those that did suffer sales losses but did not get the policy support, by 4 to 5 pp. Futhermore, digitalisation spending increases with pre-crisis productivity. This result comes after conditioning on the sector and country in which the firms operate. Hence, in the same country and sector, the more productive firms are more likely to digitalise further. This dynamic may contribute to widening the productivity gap as digitalisation is likely to foster productivity. Finally, stronger firms, i.e. those not in distress, having lower leverage or higher capital base, are more likely to digitalise. However, these effects are not significant at 10%.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Policy support	<b>0.050</b> ***	0.053***	0.055***	0.052***	0.052***	0.051***	0.052***	<b>0.047</b> ***	<b>0.048</b> ***
Sales loss	[0.010]	[0.010]	[0.011]	[0.013]	[0.011]	[0.014]	[0.013]	[0.017]	[0.014]
less than 25%		0.009	0.010	0.022	0.010	0.019	0.016	0.026	0.018
		[0.012]	[0 013]	[0.015]	[0.012]	[0.018]	[0.019]	[0.022]	[0.018]
between 25 to 50%		-0.028*	-0.021	-0.018	-0.024	-0.039	-0.039	-0.037	-0.042*
		[0.015]	[0.016]	[0.018]	[0.015]	[0.024]	[0.026]	[0.031]	[0.024]
more than 50%		-0.023	-0.025	-0.008	-0.023	-0.084**	-0.071*	-0.071	-0.083**
		[0.022]	[0.023]	[0.028]	[0.022]	[0.034]	[0.039]	[0.047]	[0.035]
Policy support X		[0:022]	[0:020]	[0:020]	[0:022]	[0105 1]	[01059]	[01017]	[0:000]
sales loss less than 25%						-0.016	-0.009	-0.005	-0.012
						[0.024]	[0.026]	[0.029]	[0.025]
sales loss bw. 25 to 50%						0.017	0.028	0.028	0.027
						[0.030]	[0.032]	[0.038]	[0.030]
sales loss more than 50%						0.082*	0.063	0.084	0.081*
						[0.042]	[0.047]	[0.057]	[0.043]
Pre-covid productivity	0.033***	0.032***	0.031***	0.027***	0.030***	0.032***	0.031***	0.027***	0.030***
1 5	[0.007]	[0.007]	[0.007]	[0.008]	[0.007]	[0.007]	[0.007]	[0.008]	[0.007]
Financial leverage			-0.001				-0.001		
5			[0.005]				[0.005]		
Firm in distress				-0.029*				-0.029*	
				[0.016]				[0.016]	
Capital ratio					0.002				0.002
1					[0.019]				[0.019]
Observations	8,819	8,819	7,792	6,089	8,541	8,819	7,792	6,089	8,541
R-squared	0.091	0.091	0.096	0.093	0.091	0.092	0.096	0.094	0.091
Firm controls	yes	yes	yes	yes	yes	yes	yes	yes	yes
Country FE	yes	yes	yes	yes	yes	yes	yes	yes	yes
Sector FF	Ves	Ves	Ves	Ves	Ves	Ves	Ves	Ves	Ves

### **Table 4** Factors explaining the likelihood of becoming more digital (firms that have taken actions or made investments to become more digital)

**Source:** Authors' estimations based on EIBIS21. **Note:** Linear Probability Model estimated with firm size dummies (small, medium and large), sectors (as in Figure 3) and firm age dummies: Less than 2 years; 2 years to less than 5 years; 5 years to less than 10 years; 10 years to less than 20 years; 20 years or more. Constant not reported. Robust standard errors in brackets, \*\*\*p < 0.01, \*\*p < 0.05, \*p < 0.1.

**Table 5** shows the results explaining the likelihood of strengthening digitalisation, augmenting the set of variables considered above with variables related to financial expansion, debt as well as equity. As in **Table 4**, in all the estimations, sales losses has a negative impact on digitalisation, reducing the likelihood to digitalise more by 5 to 10 pp. However, the effect is compensated by the policy allotment. On the one hand, firms that received policy support are unconditionally more likely to digitalise more, by 4 to 5 pp. On the other hand, firms that received the policy support while suffering large sale losses, are more likely to digitalise than those that did suffer sale losses but did not get the policy support, by 5 pp. Also as before, digitalisation spending increases with pre-crisis productivity. Finally, stronger firms, those not in distress, having lower leverage or higher capital base, are more likely to digitalise. Again, these effects are not significant at 10%.

More interestingly, in all the cases, firms that have increased their external financing are more likely to digitalise, an effect that is always significant at 10% at least. Hence, increased equity

raises the probability to digitalise by 4 to 5 pp. A similar, but slightly stronger effect is found for debt. When considered jointly with the factors contributing to explain the stronger equity base, this finding suggests that the policy support deployed at the occasion of the crisis could have fostered the crowing-in of investors and thereby speeding up the digital transformation of European firms.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	0.041+++	0.045444	0.045444	0.042444	0.044555	0.042555	0.042444	0.02044	0.040444
Policy support	0.041***	0.045***	0.047***	0.043***	0.044***	0.043***	0.043***	0.039**	0.040***
	[0.011]	[0.011]	[0.011]	[0.013]	[0.011]	[0.014]	[0.015]	[0.017]	[0.014]
Sales loss		•		•		•	-	•	•
less than 25%		0.007	0.007	0.020	0.008	0.017	0.013	0.024	0.016
		[0.012]	[0.013]	[0.015]	[0.012]	[0.018]	[0.019]	[0.022]	[0.018]
between 25 to 50%		-0.036**	-0.029*	-0.025	-0.033**	-0.047**	-0.047*	-0.043	-0.051**
		[0.015]	[0.016]	[0.018]	[0.015]	[0.024]	[0.026]	[0.031]	[0.024]
more than 50%		-0.036	-0.038	-0.019	-0.035	-0.093***	-0.081**	-0.081*	-0.091***
		[0.022]	[0.023]	[0.028]	[0.022]	[0.034]	[0.039]	[0.047]	[0.035]
Policy support X									
sales loss less than 25%						-0.015	-0.008	-0.006	-0.011
						[0.024]	[0.026]	[0.029]	[0.025]
sales loss bw. 25 to 50%						0.017	0.027	0.026	0.027
						[0.030]	[0.032]	[0.038]	[0.030]
sales loss more than 50%						0.078*	0.059	0.082	0.077*
						[0.042]	[0.047]	[0.057]	[0.043]
Debt Increase	0.034***	0.034***	0.033***	0.029***	0.032***	0.034***	0.033***	0.029***	0.031***
	[0.007]	[0.007]	[0.007]	[0.008]	[0.007]	[0.007]	[0.007]	[0.008]	[0.007]
Equity injection	0.046***	0.052***	0.054***	0.046***	0.053***	0.051***	0.054***	0.045***	0.053***
1 2 3	[0.015]	[0.015]	[0.016]	[0.017]	[0.015]	[0.015]	[0.016]	[0.017]	[0.015]
Pre-covid productivity	0.044**	0.048**	0.051**	0.053**	0.047**	0.047**	0.051**	0.053**	0.047**
	[0.021]	[0.021]	[0.023]	[0.026]	[0.021]	[0.021]	[0.023]	[0.026]	[0.021]
Financial leverage	[]	[]	-0.002	[]	[]	[]	-0.002	[]	[]
8-			[0.005]				[0 005]		
Firm in distress			[0.005]	-0 033**			[0.005]	-0 033**	
1 mm m distress				-0.035 [0.016]				-0.035	
Conital ratio				[0.010]	0.011			[0.010]	0.011
Capital fatio					0.011				0.011
					[0.019]				[0.019]
Observations	8 810	8 810	7 702	6 080	8 5/11	8 810	7 702	6 080	8 541
D sequered	0,019	0,019	1,192	0,005	0,041	0,019	0.002	0,089	0,041
K-squared	0.092	0.093	0.098	0.095	0.093	0.094	0.098	0.096	0.093
r initi controls	yes	yes	yes	yes	yes	yes	yes	yes	yes
Country FE	yes	yes	yes	yes	yes	yes	yes	yes	yes
Sector FE	Vec	Vec	Vec	Vec	Vec	Vec	Vec	Vec	Vec

Table 5	Likelihood	of becoming	more digital
			more engreen

Source: Authors' estimations based on EIBIS21. Note: Linear Probability Model estimated with firm size dummies (small, medium and large), sectors (as in Figure 3) and firm age dummies: Less than 2 years; 2 years to less than 5 years; 5 years to less than 10 years; 10 years to less than 20 years; 20 years or more. Constant not reported. Robust standard errors in brackets, \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

#### 5. Concluding remarks

In Europe, the policy support deployed during the crisis has been massive and multifaceted. Some have argued that government support is exposed to moral hazard problems: being mostly unconditional, it maintains firms that would have otherwise disappeared without the support triggered by the crisis. According to this argument, the policy deployed could lower the growth prospects of the European economy post-crisis.

Our findings do not support this view. The main explanatory factor to get the policy support was the change in sales during the crisis period and we do not find evidence of zombie lending in the policy allotment as firm weakness indicators do not appear to explain the allotment. We do find that firms with low liquidity buffers received more support and that this difference was significant. But this suggests that the first goal of the policy, avoiding a liquidity dry-out and freezing the corporate ecosystem, was achieved.

Furthermore, we find some evidence that firms that received the policy support are more positive about their investment outlook. They may have been in a better position to crowd-in investors and recapitalise. Finally, we find evidence that the conjunction of policy support and stronger equity base accelerates the digital transformation of European corporates, a transformation that the crisis has made even more necessary.

Taken together, our findings suggest several measures that policy makers might implement as a second step for a smoother economic recovery and transition to the new normal. After avoiding the liquidity crunch during the temporary halt of the economic activity in 2020, there is a clear need for new policy tools to support corporate solvency. The equity-type instruments are needed as a second step for rebalancing the balance sheet structure of corporates, affected by both strong losses and increasing indebtedness resulted also from successful liquidity support. Moreover, higher long-term growth can be achieved with common public-private efforts in direction of adopting new digital solutions. While corporates are increasingly in favour of digital investments in a post-pandemic world, technological readiness in terms of digital infrastructure and skills are crucial, also to avoid widening cross-regional gaps. Targeted grants might further accelerate digital investments on the firm level.

#### References

- Altavilla, C., Barbiero, F., Boucinha, M. and Burlon, L. (2020). "The great lockdown: Pandemic response policies and bank lending conditions." CEPR DP No. 15298, September 2020.
- Andersson, M., L. Maurin and D. Rusinova, (2021). "Market finance as a spare tire? Corporate investment and access to bank credit in Europe", EIB Working Paper Series.

- Andreeva, D. and García-Posada, M. (2020). "The Impact of the ECB's Targeted Long-Term Refinancing Operations on Banks' Lending Policies: The Role of Competition," ECB Working Paper, 2364.
- Archarya V.V. and S. Steffen, (2020). "The risk of being a fallen angel and the corporate sector dash for cash in the midst of Covid", NBER, WP 27601, July.
- Asdrubali, P. and Signore, S. (2015). "*The Economic Impact of EU Guarantees on Credit to SMEs – Evidence from CESEE Countries*," EIF Working Paper Series 2015/29, European Investment Fund (EIF).
- Autor, D., Cho, D., Crane, L. D., Goldar, M., Lutz, B., Montes, J., Peterman, W. B., Ratner, D., Villar, D. and Yildirmaz, A. (2020). "An Evaluation of the Paycheck Protection Program Using Administrative Payroll Microdata," mimeo.
- Banerjee R. and B. Hofmann, (2020). "Corporate zombies: Anatomy and life cycle", BIS Working Papers, n° 882.
- Barnes, S., Hillman, R., Wharf, G. and MacDonald, D. (2021). "The impact of COVID-19 on corporate fragility in the United Kingdom: Insights from a new calibrated firm-level Corporate Sector Agent-Based (CAB) Model," OECD Economics Department Working Papers 1674, Organisation for Economic Co-operation and Development (OECD).
- Bertoni, F., Brault, J., Colombo, M. G., Quas, A. and Signore, S. (2019). "*Econometric study* on the impact of EU loan guarantee financial instruments on growth and jobs of SMEs," EIF Working Paper Series 2019/54, European Investment Fund (EIF).
- Bertoni, F., Colombo, M. G. and Quas, A. (2018). "The effects of EU-funded guarantee instruments of the performance of small and medium enterprises: Evidence from France," EIF Working Paper Series 2018/52, European Investment Fund (EIF).
- Betz, F, and De Santis, R. A. (2021), "ECB corporate QE and the loan supply to bank-dependent firms," *International Journal of Central Banking*, forthcoming.
- Bighelli, T., Lalinksy, T. and CompNet Data Providers (2021). "COVID-19 government support and productivity: Micro-based cross-country evidence," CompNET Policy Brief 14, CompNET.
- Blanco, R., Mayordomo, S., Menéndez, Á and Mulino, M. (2021). "Impact of the COVID-19 crisis on Spanish firms' financial vulnerability," BdE Documentos Ocasionales 2119, Banco de España (BdE).
- Brault, J. and Signore, S. (2019). "*The real effects of EU loan guarantee schemes for SMEs: A pan-European assessment*," EIF Working Paper Series 2019/56, European Investment Fund (EIF).
- Brown, J. D. and Earle, J. S. (2017). "Finance and Growth at the Firm Level: Evidence from SBA Loans," *Journal of Finance*, 72(3): 1039-1080.
- Brutscher, P-B., Coali, A., Delanote, J., Harasztosi, P., 2020. "EIB Group Survey on Investment and Investment Finance: A technical note on data quality," EIB Working Papers 2020/08, European Investment Bank (EIB).
- Bureau, B., Duquerroy, A., Giorgi, J., Lé, M., Scott, S. and Vinas, F. (2021). "L'impact de la crise sanitaire sur la situation financière des entreprises en 2020 : une analyse sur données individuelles," Banque de France Working Paper 824, Banque de France.

- Carletti, E., Oliviero, T., Pagano, M., Pelizzon L. and Marti G. Subrahmanyam (2020), "The COVID-19 Shock and Equity Shortfall: Firm-level Evidence from Italy". *The Review of Corporate Finance Studies*, Vol 9, Issue 3, pp 534–568
- Core, F. and De Marco, F. (2021). "Public Guarantees for Small Businesses in Italy during Covid-19," CEPR Discussion Papers 15799, Centre for Economic Policy Research (CEPR).
- Degryse, H. and Ongena, S. (2005). "Distance, Lending Relationships, and Competition," *Journal of Finance*, 60(1): 231–266.
- Demmou, L., Franco, G., Calligaris, S. and Dlugosch, D. (2021). "Liquidity shortfalls during the COVID-19 outbreak: Assessment and policy responses," OECD Economics Department Working Papers 1647, Organisation for Economic Co-operation and Development (OECD).
- Demmou, L., Calligaris, S., Franco, G., Dlugosch, D., Adelat McGowan, M. and Sakha, S. (2021). "Insolvency and debt overhang following the COVID-19 outbreak: assessment of risks and policy responses," OECD Economics Department Working Papers 1651, Organisation for Economic Co-operation and Development (OECD).
- De Santis, R. A., Ferrando, A., and Gabbani, E. S. (2021). "The impact of fiscal support measures on the liquidity needs of firms during the pandemic," ECB Economic Bulletin, Issue 4/2021, European Central Bank (ECB).
- Díez, F. J., Duval, R., Fan, J., Garrido, J., Kalemli-Özcan, S., Maggi, C., Martinez-Peria, S. and Pierri, N. (2021). "Insolvency Prospects Among Small and Medium Enterprises in Advanced Economies: Assessment and Policy Options," IMF Staff Discussion Note 2021/002, International Monetary Fund (IMF).
- Simeon Djankov S., Eva (Yiwen) Zhang E. (2021) "As COVID rages, bankruptcy cases fall" VoxEU.org, 04 February.
- D'Ignazio, A. and Menon, C. (2020). "Causal Effect of Credit Guarantees for Small- and Medium-Sized Enterprises: Evidence from Italy," *Scandinavian Journal of Economics*, 122(1): 191-218.
- Doniger, C. L. and Kay, B. (2021). "Ten Days Late and Billions of Dollars Short: The Employment Effects of Delays in Paycheck Protection Program Financing," Finance and Economics Discussion Series 2021-003, Board of Governors of the Federal Reserve System.
- Drabancz, Á, El-Meouch, N. M. and Lang, P. (2021). "A koronavírus-járvány miatt bevezetett jegybanki és állami hitelprogramok hatása a magyar foglalkoztatásra," Közgazdasági Szemle, 68(September):930-965.
- Dvouletý, O., Čadil, J. and Mirošní, K. (2019). "Do Firms Supported by Credit Guarantee Schemes Report Better Financial Results 2 Years After the End of Intervention?," The B.E. Journal of Economic Analysis & Policy, 19(1):1-20.
- Ebeke, C., Jovanovic, N., Valderrama, L. and Zhou, J. (2021). "Corporate Liquidity and Solvency in Europe during COVID-19: The Role of Policies," IMF Working Paper 21/56, International Monetary Fund (IMF).
- EIB (2021), "Financing corporate investment", Chapter 3, Investment Report, forthcoming.
- Endresz, M., Harasztosi P. and R. P. Lieli, 2015. "The Impact of the Magyar Nemzeti Bank's Funding for Growth Scheme on Firm Level Investment," MNB Working Papers 2015/2, Magyar Nemzeti Bank (Central Bank of Hungary).
- Erhardt, E. C. (2017). "Microfinance beyond self-employment: Evidence for firms in Bulgaria," *Labour Economics*, 47(C): 75-95.

- Faulkender, M., Jackman, R. and Miran, S. I. (2021). "The Job-Preservation Effects of Paycheck Protection Program Loans," Office of Economic Policy Working Paper 2020-01, Department of the Treasury.
- Ferrando, A., Popov, A. and Udell, G. F. (2019). "Do SMEs Benefit from Unconventional Monetary Policy and How? Microevidence from the Eurozone," *Journal of Money, Credit and Banking*, 51(4): 896-928.
- France Stratégie (2021a). "Comité de suivi et d'évaluation des mesures de soutien financier aux entreprises confrontés à l'épidémie de Covid-19 Rapport final," Inspection générale des finances & France Stratégie, présidé par Benoît Coeuré.
- France Stratégie (2021b). "Committee on the monitoring and evaluation of financial support measures for companies confronted with the Covid-19 epidemic Progress report», ", Rapport final, Inspection générale des finances & France Stratégie, présidé par Benoît Coeuré.
- Gourinchas, P.-O., Kalemli-Ozcan, S., Penciakova, V. and Sander, N. (2020). "COVID-19 and SME Failures," CEPR Discussion Papers 15323, Centre for Economic Policy Research (CEPR).
- Gourinchas, P.-O., Kalemli-Ozcan, S., Penciakova, V. and Sander, N. (2021). "Fiscal Policy in the Age of COVID: Does it 'Get in all of the Cracks?", NBER Working Paper 29293, National Bureau of Economic Research (NBER).
- Granja, J., Makridis, C., Yannelis, C. and Zwick, E. (2020). "Did the Paycheck Protection Program Hit the Target", NBER Working Paper 27095, National Bureau of Economic Research (NBER).
- Guipponi, G. and Landais, C. (2020). "Subsidizing Labor Hoarding in Recessions: The Employment & Welfare Effects of Short-Time Work," CEPR Discussion Papers 13310, Centre for Economic Policy Research (CEPR).
- Hadjibeyli, B. Roulleau, G. and Bauer, A. (2021). "Live and (don't) let die: the impact of COVID-19 and public support on French firms," Direction générale du Trésor Working Papers 2021/2, Ministère de l'Économie, des Finances et de la Relance.
- Hoffman, M. and Schneck, S. (2011). "Short-Time Work in German Firms," Applied Economics Quarterly, 57(4): 233-254.
- Horvath, A. and Lang, P. (2021). "Do loan subsidies boost the real activity of small firms?," *Journal of Banking & Finance*, 122(C), 105988.
- Hubbard, R. G. and Strain, M. R. (2020). "Has the Paycheck Protection Program Succeeded?," NBER Working Paper 28032, National Bureau of Economic Research (NBER).
- Kaniovski, S., Pekanov, A., and Url, T. (2021). "Ex-post-Analyse der Wirkungen des COVID-19-Maßnahmenpaketes auf die Unternehmensliquidität," Österreichisches Institut für Wirtschaftsforschung (WIFO).
- Kanngieser, D., Martin, R., Maurin, L. and D. Moccero, (2020), "The macroeconomic impact of shocks to bank capital buffers in the Euro Area", *B.E. Journal of Macroeconomics*, Vol. 20, Issue 1, 17.
- Kapinos, P. (2021a). "Paycheck Protection Program: County-Level Determinants and Effect on Unemployment," Federal Reserve Bank of Dallas Working Paper 2105, Federal Reserve Bank of Dallas.

- Kapinos, P. (2021b). "Did the Paycheck Protection Program have negative side effects on small-business activity?," *Economics Letters*, 208(November): 110045.
- Kopp, D. and Siegenthaler, M. (2019). "Short-Time Work and Unemployment in and after the Great Recession," KOF Working Papers 462, KOF Swiss Economic Institute, ETH Zürich.
- Lalinsky, T. and Pal, R. (2021). "Efficiency and effectiveness of the COVID-19 government support: Evidence from firm-level data," EIB Working Papers 2021/06, European Investment Bank (EIB).
- Lopez, J. A. and Speigel, M. M. (2021). "Small Business Lending Under the PPP and PPPLF Programs," Federal Reserve Bank of San Francisco Working Paper 2021-10, Federal Reserve Bank of San Francisco.
- Lopez-Garcia, P. (2020). "Box 2 The impact on potential output of a surge in firm exits as a result of COVID-19" in eds. Bodnár, K., Le Roux, J., Lopez-Garcia, P. and Szörfi, B. "The impact of COVID-19 on potential output in the euro area," ECB Economic Bulletin, Issue 7/2020, European Central Bank (ECB).
- Lydon, R., Mathä, T. Y., and Millard, S. (2019). "Short-time work in the Great Recession: firm-level evidence from 20 EU countries," IZA Journal of Labour Policy, 8(2): 1-29.
- Martín-García, R. and Morán Santor, J. (2021). "Public guarantees: a countercyclical instrument for SME growth. Evidence from the Spanish Region of Madrid," Small Business Economics, 56(1): 427-449.
- Maurin L. and R. Pal, (2020). "Investment vs debt trade-offs in the post-COVID-19 European economy", EIB Working Paper Series, 2020/09.
- Portes J. (2020) " The lasting scars of the Covid-19 crisis: Channels and impacts ll" VoxEU.org, 01 June.
- Schivardi, F., Sette, E., and Tabellini, G. (2020). "Identifying the Real Effects of Zombie Lending," The Review of Corporate Finance Studies, 9(3): 569–592.

## **APPENDIX** - Predicted probability of getting specific policy support across firm characteristics pre-COVID-19 for each type of policy separately



Subsidized of guaranteed credit (predicted probability, %)

Payments Deferrals (predicted probability, %)





Subsidies or any other type of financial support (predicted probability, %)

**Source:** Authors' estimations based on the EIBIS2021 matched with the ORBIS database. **Note:** The vertical line reports the 95% interval confidence of the conditional probability of getting the support (see EQ2). Two overlapping lines indicate that the factor does not alter significantly the probability. Red bars indicate when the characteristic is statistically discriminant at the 95% confidence level.

ECONOMICS - WORKING PAPERS 2022/01

## Firm-level policy support during the crisis:

So far, so good?

European Investment Bank The EU bank

Economics Department economics@eib.org www.eib.org/economics

European Investment Bank 98-100, boulevard Konrad Adenauer L-2950 Luxembourg +352 4379-22000 www.eib.org – info@eib.org

© European Investment Bank, 01/2022 EN

PDF: ISBN 978-92-861-5210-8